Author's response to reviews

Title: Successful management of refractory pleural effusion due to systemic AL amyloidosis by vincristine adriamycin dexamethasone chemotherapy: a case report

Authors:

Toshikazu Araoka (andy7andy14@gmail.com)
Hiroya Takeoka (htakeoka@ares.eonet.ne.jp)
Keisuke Nishioka (nishioka@kuhp.kyoto-u.ac.jp)
Masaki Ikeda (ghiblist@hotmail.com)
Makiko Kondo (maki_maki716@hotmail.com)
Azusa Hoshina (azukozu@hotmail.com)
Seiji Kishi (nephro.endoscopy@gmail.com)
Makoto Araki (makaraki@kuhp.kyoto-u.ac.jp)
Rokuro Mimura (andy14@goo.jp)
Taichi Murakami (tatsunoya@clin.med.tokushima-u.ac.jp)
Akira Mima (amima423@gmail.com)
Kojiro Nagai (knagai@clin.med.tokushima-u.ac.jp)
Hideharu Abe (abeabe@clin.med.tokushima-u.ac.jp)
Toshio Doi (doi@clin.med.tokushima-u.ac.jp)

Version: 3 Date: 26 March 2010

Author's response to reviews:

Mar. 26th, 2010

The Editor
Journal of Medical Case Reports

re: “Successful management of refractory pleural effusion due to systemic AL amyloidosis by vincristine adriamycin dexamethasone chemotherapy: a case report” by Toshikazu Araoka, Hiroya Takeoka, Keisuke Nishioka, Masaki Ikeda, Makiko Kondo, Azusa Hoshina, Seiji Kishi, Makoto Araki, Rokuro Mimura, Hideharu Abe and Toshio Doi; MS: 1956480305333764

Dear Editor:

We are most grateful to you for the helpful comments on the revision version of our manuscript.

In the attached document, we describe the changes made in response to your major and minor comment point by point.

In the revised version of our paper, we have removed any qualifications of our authors. Further, we have changed E-mail address lists, which are written as initial and E-mail addresses on the page 1 in the title page. Moreover, we have...
changed the order of our author from “Rokuro Mimura, Seiji Kishi, Makoto Araki” to “Seiji Kishi, Makoto Araki, Rokuro Mimura”.

We have changed the figure citation from “Figure A and B” to “Figures 1A and 1B” on the page 6 and from “Figure C and D” to “Figures 1C and 1D” on the page 7 in the case presentation section. Further, we have changed from “Figure” to “Figure 1” on the page 12 in the figure legends section.

In the image file, we have removed figure number and title. Further, we have cropped the white space around the image.

In the title page, we have added 3 authors because they contribute to the discussion of the revised version of our manuscript. Hence, we have added the sentence “TM, AM and KN contributed to the discussion of the revised version of the manuscript.” on the page 11 in the authors’ contributions section.

We hope that the revised version of our paper is now suitable for publication in the Journal of Medical Case Reports and we look forward to hearing from you at your earliest convenience.

Best regards.

Sincerely,

Hiroya Takeoka, M.D., Ph.D.
Division of Nephrology, Hyogo Prefectural Amagasaki Hospital, Hyogo, Japan
1-1-1 Higashidaimotsu-cho, Hyogo, 660-0828, Japan
tel:+81-6-6482-1521, fax:+81-6-6482-7430
e-mail: htakeoka@ares.eonet.ne.jp